Author's response to reviews

**Title:** Oral rehydration versus intravenous therapy for treating dehydration due to gastroenteritis in children: a meta-analysis of randomised controlled trials

**Authors:**

Steven Bellemare (sb6@ualberta.ca)
Ms Lisa Hartling (hartling@ualberta.ca)
Ms Natasha Wiebe (nwiebe@ualberta.ca)
Kelly Russell (krussell@ualberta.ca)
William R Craig (w craig@cha.ab.ca)
Don McConnell (DMcConne@cha.ab.ca)
Dr Terry P Klassen (terry.klassen@ualberta.ca)

**Version:** 4  **Date:** 17 Mar 2004

The manuscript has been re-formatted according to the instructions that were emailed to us. There were no additional compulsory revisions requested by the peer-reviewers. In response to Dr. Koretz’ second comment, we have added a sentence in the methods section (paragraph 3) and a footnote to table 2 to clarify that quality assessment was based on the information provided in the published manuscript.